Author’s response to reviews

Title: Data Access Committees

Authors:

Phaikyeong Cheah (phaikyeong@tropmedres.ac)
Jan Piasecki (jan.piasecki@uj.edu.pl)

Version: 2 Date: 27 Jan 2020

Author’s response to reviews:

27 Jan 2020

Dear Editor
Response to reviewers’ and editorial comments
Title: Data Access Committees
Authors: Phaik Yeong Cheah and Jan Piasecki

Thank you for your kind reply. Please see point by point response to the reviewer’s/editorial comments.

1. Please also state the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, in the Funding section.
   Done. See page 20 under “Funding”

2. Please format your main text and the abstract to include the following subheadings: Background, Main text, and Conclusions.
   Done in both abstract and main text.

3. Please be sure to address reviewer 2’s comments at the end of this letter.

Page 14 Line 21 "both regulations grant several derogation for data-driven research involving public benefit, posing minimal risk and allow access to IDENTIFIABLE health information."

Changed “identified” to “identifiable”. See page 7, 1st paragraph, last sentence, line 26.

We hope that our paper is now suitable for publication.

Many thanks.

Phaik Yeong Cheah
Corresponding Author